



Article Intelligent Decision Support for Energy Management: A Methodology for Tailored Explainability of Artificial Intelligence Analytics

Dimitrios P. Panagoulias ^{1,*}, Elissaios Sarmas ², Vangelis Marinakis ², Maria Virvou ¹, George A. Tsihrintzis ¹, and Haris Doukas ²

- ¹ Department of Informatics, University of Piraeus, 80 Karaoli ke Dimitriou ST, 18534 Piraeus, Greece; mvirvou@unipi.gr (M.V.); geoatsi@unipi.gr (G.A.T.)
- ² Decision Support Systems Laboratory, School of Electrical & Computer Engineering, National Technical University of Athens, 15780 Athens, Greece; esarmas@epu.ntua.gr (E.S.); vmarinakis@epu.ntua.gr (V.M.); h_doukas@epu.ntua.gr (H.D.)
- Correspondence: panagoulias_d@unipi.gr

Abstract: This paper presents a novel development methodology for artificial intelligence (AI) analytics in energy management that focuses on tailored explainability to overcome the "black box" issue associated with AI analytics. Our approach addresses the fact that any given analytic service is to be used by different stakeholders, with different backgrounds, preferences, abilities, skills, and goals. Our methodology is aligned with the explainable artificial intelligence (XAI) paradigm and aims to enhance the interpretability of AI-empowered decision support systems (DSSs). Specifically, a clustering-based approach is adopted to customize the depth of explainability based on the specific needs of different user groups. This approach improves the accuracy and effectiveness of energy management analytics while promoting transparency and trust in the decision-making process. The methodology is structured around an iterative development lifecycle for an intelligent decision support system and includes several steps, such as stakeholder identification, an empirical study on usability and explainability, user clustering analysis, and the implementation of an XAI framework. The XAI framework comprises XAI clusters and local and global XAI, which facilitate higher adoption rates of the AI system and ensure responsible and safe deployment. The methodology is tested on a stacked neural network for an analytics service, which estimates energy savings from renovations, and aims to increase adoption rates and benefit the circular economy.

Keywords: machine learning; optimization; explainable artificial intelligence; energy management; energy transition; circular economy

1. Introduction

The energy sector undergoes significant transformations that impact all its stakeholders, collectively known as "Energy Transition". This term refers to the shift from fossil-fuel-based energy production, such as oil, natural gas, or coal, to more sustainable and eco-friendly sources like solar energy, wind energy, hydrogen-based energy, or the use of lithium-ion batteries. Indeed, since the Paris Agreement [1,2], there is increasing international concern and action taken with regard to climate change and the viability of Earth. More specifically, the Paris Agreement focuses on the design and application of viable, effective, socially acceptable, and fair policies to fight and possibly reverse climate change on a global level [3].

Energy transition is not restricted to decarbonization only, but it also incorporates multiple social, technological, and environmental targets [4]. The concept is based on four fundamental principles, known as the 4Ds of the energy sector, which include decarbonization, digitization, decentralization, and democratization.



Citation: Panagoulias, D.P.; Sarmas, E.; Marinakis, V.; Virvou, M.; Tsihrintzis, G.A.; Doukas, H. Intelligent Decision Support for Energy Management: A Methodology for Tailored Explainability of Artificial Intelligence Analytics. *Electronics* 2023, 12, 4430. https://doi.org/ 10.3390/electronics12214430

Academic Editor: Ahmed Abu-Siada

Received: 22 September 2023 Revised: 12 October 2023 Accepted: 24 October 2023 Published: 27 October 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In more detail, decarbonization is a key pillar of energy transition [5], referring to the reduction in dependence on carbon for energy production and its gradual replacement with other sources such as renewable energy sources (RESs) [6]. The ultimate goal of decarbonization is a world economy that does not produce CO₂ emissions. Decentralization is another pillar of energy transition, focusing on the construction of energy production systems near energy consumers, reducing the dependence on fossil fuels and promoting the use of RESs [7].

The success of decentralization depends on the existence of active consumers, also known as "prosumers," who produce, store, and manage the energy that they consume [8]. Democratization follows decentralization, as prosumers and small and medium-sized enterprises gain reliable access to low-cost energy [9]. Digitization is the fourth pillar, involving the use of electronic tools, systems, and devices to process data and extract meaningful information to optimize energy processes [10,11], as energy data exhibit the so-called 5Vs, i.e., very high volume, variety, velocity, veracity, and value [12,13].

The shift from a producer/provider-centered energy system to a consumer/householdcentered system poses significant challenges [14]. Fortunately, advancements in artificial intelligence (AI) have provided a powerful tool to facilitate this transition and achieve associated ecological and social objectives [15]. However, transitioning to autonomous AIempowered systems requires addressing their impact on stakeholders and understanding how the underlying functions achieve prediction goals [16]. It is essential for all energy stakeholders to be actively involved in decision making regarding the production, distribution, and management of energy. Stakeholder-tailored decision support systems (DSSs) that provide decision support at various levels and forms need to be developed.

These DSSs need to integrate predictive models and optimization algorithms that rely on state-of-the-art machine learning (ML) and AI technologies. Additionally, they need to incorporate explainable and responsible AI (XAI and RAI, respectively) technologies to be adopted by various user classes and to exclude unacceptable decisions. XAI and RAI technologies provide system users with justification for recommended actions [17,18]. As the explainability and responsibility of an AI system are concepts related to a particular audience, i.e., to a class of users or even specific individuals [17], the identification of various stakeholder classes is necessary.

The pivotal attributes of explainable artificial intelligence (XAI) include fairness, ethics, transparency, privacy, security, accountability, and safety. System transparency is imperative, equally weighed with the demand for data privacy and security [19]. Ensuring users comprehend the characteristics of utilized models and providing a transparent representation of the deployed algorithms are crucial [17]. The predominant issue pertains to establishing user trust in machine learning (ML) models for predictions and decision support, while developers must furnish adequate descriptions and decision-making roadmaps [20].

XAI is deemed vital due to the growing social impact of intelligent systems. Indeed, XAI promotes system adoption and could integrate familiarity with technology and AI literacy into its framework and design process, combining scientific and simplified explanations per user comprehension and mental capacity [21]. The concept of XAI is divided into global explainability, providing overarching understanding of model predictions, and local explainability, elucidating individual classifier decisions, Both global and local explainability are pivotal for assessing AI system fairness and transparency [22]. Further divisions of explainability can be assumed with regard to model architecture, benchmarks, and semantic grouping. All of these themes are examined in the following sections.

In view of the above, this paper presents a novel methodology that focuses on developing *customized* XAI analytics for energy management, which are specifically tailored to meet the needs of various groups of energy stakeholders. This methodology aims to resolve the issue of the "black box" phenomenon associated with AI decision making by ensuring the transparency and interpretability of the analytics results. To support this methodology, the paper introduces an intelligent decision support system (IDSS) that constitutes a software tool consisting of energy services. Specifically, the IDSS is developed using an iterative agile life cycle, which is a method for software development that breaks down the work into smaller sprints. The IDSS uses AI and optimization algorithms to provide optimal solutions for energy management and efficiency in buildings under specific constraints. The aim of the IDSS is to help energy stakeholders make informed decisions about energy management and efficiency in buildings by combining AI and optimization algorithms.

The methodology for developing the XAI analytics involves several processes, starting from identifying key stakeholders in the energy sector and conducting an empirical study to determine the usability and explainability needs of the IDSS. This includes creating an appropriate questionnaire, conducting a survey among key stakeholders, and performing a user clustering analysis to define the necessary depth of AI explainability in terms of AI literacy score, acceptance, and perceived usefulness.

Finally, an XAI framework is implemented based on the outcomes of the analysis, which includes XAI clusters and local and global XAI. This framework is designed to ensure a responsible and safe deployment of the AI system, while enabling higher adoption rates by promoting trust and reliability in the analytics results.

The ultimate goal of this methodology is to provide a cutting-edge solution for decision support analytics for energy management and efficiency in buildings that is trustworthy, reliable, and meets the specific needs of energy stakeholders.

In summary, the key contributions of this paper are the following:

- Agile development and implementation of an IDSS, prioritizing XAI tailored to specific user needs.
- Testing of the methodology on a stacked neural network.
- Emphasis on explainability not only as an ethical approach but also as a crucial tool
 for building user trust and ensuring rapid adoption rates in the energy sector, thus
 contributing to a more sustainable future.
- Utilization of the technology acceptance model (TAM) to cluster stakeholders based on their perceptions of usefulness and ease of use, where:
 - Perception of usefulness is associated with AI literacy.
 - Perceived ease of use is linked with usability.
- Identification that varying perceptions necessitate different explanation approaches, therefore grouping XAI tools accordingly to ensure broader and accelerated adoption rates.

The main body of the paper is structured as follows: Section 2 discusses energy-related issues and introduces the IDSS developed through an agile process. Section 3 introduces a novel methodology for tailored AI explainability in energy management IDSSs. Section 4 identifies and discusses key energy sector stakeholders. Section 5 describes the creation of a stakeholder questionnaire, designed considering AI explainability, energy market trends, and the TAM. Section 6 presents the empirical study and user-based clustering analytics conducted through an actual survey of stakeholders, which specifies requirements for the tailored explainability of energy DSS analytics taking into account AI literacy and levels of abstraction. Section 7 describes the actual implementation and testing of personalized/tailored XAI in the IDSS, highlighting its utility and adoption implications. Lastly, Section 8 summarizes the findings and conclusions and proposes future research directions.

2. Energy Management IDSS with Analytics Services Using an Agile Development Process

The energy sector is navigating through three primary challenges: (1) management and optimization of renewable energy sources, (2) handling distributed energy resources, and (3) enhancing building energy efficiency [23,24]. Managing distributed energy resources involves addressing issues like predicting energy demand and optimizing loads and energy storage [25–30]. Additionally, the circular economy model seeks to intertwine sustainability with economic growth through strategies focusing on resource reusability and recycling

and backed by regulatory frameworks and incentives [31–35]. ML and AI present robust avenues to address these multifaceted challenges in the energy sector.

2.1. Agile Software Life Cycle in Sprints of Energy Management IDSS

The development of analytics services for a model energy management intelligent decision support system (IDSS) has previously progressed iteratively within an agile life cycle, incorporating sprints illustrated in Figure 1. The agile methodology emphasizes continuous delivery and iterative development, and encompasses stages such as planning, design, development, testing, deployment, revising, and releasing, all within time-bound sprints, each yielding a functional software increment [36–38]. The energy management IDSS, utilized as a research testbed, comprises eight analytics services, each constructed in a sprint within the agile framework.



Figure 1. Agile life cycle of energy management IDSS consisting of 8 analytics services released in sprints.

2.2. Analytics Services for Energy Management IDSS

In this section, eight analytical services (ASs) [39–41] constituting the energy management intelligent decision support system (IDSS) are briefly described, with their interconnection depicted in Figure 2.



Figure 2. AI/ML/optimization-based analytics services.

1. AS01: Focuses on photovoltaic (PV) production forecasting using PV production time series and weather data to estimate short- to mid-term PV system energy generation.

- 2. AS02: Addresses building-level consumption forecasting, utilizing building consumption time series to estimate future energy use.
- AS03: Concentrates on grid load forecasting, using building consumption and weather data to predict future grid loads.
- 4. AS04: Estimates energy savings from renovations, utilizing pre- and post-renovation consumption time series of the building, weather data, and measurements.
- AS05: Pertains to financing energy conservation measures, using historical project records, building characteristics, and cost data, outputting a renovation class (A, B, or C) indicating potential against cost.
- AS06: Optimizes distributed energy resources in micro-grids, using forecasts and profiles from PV production, building consumption, loads, and storage systems to output optimal scheduling and sizing, interlinking with AS01 and AS02.
- 7. AS07: Involves flexible load scheduling for peak shaving at the grid level, using forecasts and profiles from PV production and grid load, interlinking with AS01 and AS03.
- 8. AS08: Manages thermal comfort with dynamic energy management, utilizing roomlevel consumption, humidity, and temperature data to schedule HVAC systems, connecting directly with AS02.

Later on, in Figure 5, the interconnections between the defined analytics services and key stakeholders of the energy sector (identified in Section 4) will be illustrated.

3. Overview of the Novel Methodology for Tailored Explainability of AI in Energy Management IDSS

Toward facilitating the adoption of IDSS in energy management applications, the analytic services that they provide need to be more understandable and comprehensible by each associated stakeholder. The methodology described in this paper offers a novel approach to designing a tailored explainability solution for AI analytics services in the energy management sector. It incorporates multiple processes that are informed by the stakeholders' specific needs and preferences, and it applies established frameworks such as the TAM and human–computer interaction (HCI) principles to ensure the effectiveness of the solution.

One of the key contributions of this methodology is that it takes into account the diverse needs and preferences of stakeholders in the energy management sector. By conducting a survey and analyzing the data, the methodology is able to identify the different clusters of stakeholders with varying levels of AI literacy, acceptance, perceived usefulness and ease of use, and energy market trends. This helps to ensure that the tailored explainability solution is designed to meet the specific needs of each stakeholder group.

Another contribution of this methodology is that it incorporates both local and global levels of implementation. This ensures that the tailored explainability solution is accessible to all stakeholders who require it. This is particularly important in the energy management sector, where stakeholders may vary significantly in their backgrounds and needs.

In general, the novelty and contribution of this methodology lie in its comprehensive approach to designing a tailored explainability solution for AI analytics services in the energy management sector. By taking into account the stakeholders' specific needs and preferences and applying established frameworks, the methodology is able to develop an effective solution that meets the diverse needs of stakeholders in this sector.

In more detail, the methodology consists of five processes aimed at developing a tailored explainability solution for AI analytics services in the energy management sector, as follows.

Process 1: Identification of key stakeholders and AI analytics services. The first process involves identifying the key stakeholders in the energy management sector and the required AI analytics services. This step is critical because it helps to understand the ecosystem in which the AI analytics services will be deployed and identify the stakeholders who will be impacted by these services. By understanding the interconnection between the stakeholders and the analytics services, it becomes possible to design a tailored explainability solution that addresses their specific needs.

Process 2: Creation of tailored explainability requirements questionnaire. The second process involves creating a tailored questionnaire to gather requirements for explainability. This questionnaire is designed to capture the stakeholders' perceptions of the ease of use and usability of the AI analytics services, their level of AI literacy, and their understanding of the energy market trends. The questionnaire applies the TAM and HCI principles, which help to ensure that the stakeholders' needs are fully addressed.

Process 3: Conducting a survey among energy management stakeholders. The third process involves conducting a survey among the energy management stakeholders to collect data on their demographics and relation to the energy sector. The survey is designed to ensure that the tailored explainability solution is informed by the specific needs of the stakeholders. This step also helps to ensure that the stakeholders are involved in the process and that their voices are heard.

Process 4: User-based clustering analysis to determine the required depth of explainability. The fourth process involves user-based clustering analysis to determine the required depth of explainability. This is carried out in terms of the stakeholders' AI literacy, acceptance, perceived usefulness and ease of use, and energy market trends. By analyzing the survey data, it becomes possible to group the stakeholders based on their needs and preferences. This step helps to ensure that the tailored explainability solution is effective in providing the required level of transparency and understanding.

Process 5: Implementation of tailored XAI. The final process involves implementing the tailored explainability solution for AI analytics services at both the local and global levels. This step involves developing a solution that meets the specific needs of the stakeholders and is effective in providing the required level of transparency and understanding. The solution is implemented at both the local and global levels to ensure that it is accessible to all stakeholders who require it.

This methodology, which is summarized in Figure 3, is designed to ensure that the explainability of AI analytics services in the energy management sector is tailored to the specific needs of stakeholders and is effective in providing the required level of transparency and understanding. It is a comprehensive approach that involves multiple steps and is informed by data collected from the stakeholders themselves.



Figure 3. The five processes and subprocesses of the methodology for tailored explainability in energy management IDSS.

In the energy sector, there are various stakeholders with different roles and responsibilities, and each stakeholder group has specific needs and requirements. Therefore, it is essential to identify these stakeholders and understand their interconnection to effectively implement any energy management system.

However, identifying stakeholders can be challenging as they can vary depending on the context, location, and type of project. Additionally, stakeholders may have conflicting interests, which can further complicate the process. Hence, it requires a systematic and comprehensive approach to identify all stakeholders and understand their requirements and needs.

It is important to note that stakeholders range from energy producers, suppliers, balance responsible parties, transmission and distribution system operators, regulatory bodies, electricity customers, consumers, residents of buildings, project developers, investing funds, governmental institutions, aggregators, facility managers, and energy managers. Each stakeholder group plays a significant role in the energy sector and requires different types of analytics services to meet their needs.

In this subsection, we outline eleven (11) key stakeholders, which are associated with stakeholders and have been fully implemented. Figure 4 illustrates key energy stakeholders along with their corresponding importance and Figure 5 illustrates the interconnection between the key stakeholders and the analytics services defined previously. The description and the roles of these stakeholders are analyzed as follows:

- 1. **KS01—Producers (PDs)**: Producers generate energy, either conventionally or through renewable sources, and include entities like individual park owners and aggregators of several production sites.
- KS02—Suppliers (SPs): Suppliers sell energy to various consumers and may not necessarily be producers, especially in competitive energy markets.
- 3. **KS03—Balance Responsible Party (BRP)**: BRPs oversee the balance of one or more access points to the transmission grid.
- 4. **KS04—Transmission System Operator (TSO)**: TSOs transport electricity from producers to distribution networks or direct to consumers and are obliged to ensure the overall balance of the network.
- KS05—Distribution System Operators (DSOs): DSOs handle the construction and management of medium- and low-voltage/pressure networks, connecting transmission networks to private properties.
- 6. **KS06—Regulatory Bodies (REGs)**: These bodies regulate the energy market, ensuring transparency and competition, defending consumer interests, and certifying energy markets' operations.
- KS07—Electricity Customers, Consumers, Residents of Buildings, Occupants (BUIs): This group consists of entities consuming energy at various scales.
- 8. **KS08—Project Developers, Investing Funds, Governmental Institutions (INVs)**: Entities focusing on financing opportunities in energy efficiency projects aiming to reduce carbon emissions at various scales.
- 9. **KS09—Aggregators (AGGs)**: Aggregators manage the electricity consumption of a consumer group, responding to total grid demand, and sell flexibility to DSOs, BRPs, and TSOs.
- 10. **KS10—Facility Managers, Building Operators (FMBs)**: FMBs manage the heating, cooling, and mechanical equipment of a building to ensure efficient operation.
- 11. **KS11—Energy Managers (ENMs)**: ENMs manage the HVAC systems of large buildings to maintain acceptable thermal comfort levels.



Figure 4. Key energy stakeholders.



Figure 5. Interconnection between key stakeholders and analytics services.

5. TAM for Tailored Questionnaire in an Empirical Study

The next step in the process entails the development of a customized empirical study aimed at obtaining information on explainability requirements. The empirical study is specifically designed to elicit feedback from stakeholders on the user-friendliness and functionality of the AI analytics services, their level of familiarity with AI, and their grasp of current energy market trends. Therefore, the empirical study is based on a survey with a questionnaire that has been created to serve the purposes of the empirical study. As such, the questionnaire is grounded on the principles of the TAM and HCI, which serve to guarantee that the stakeholders' needs are comprehensively met.

According to the TAM developed by Davis in 1989, the acceptance of a new computer system is linked to two external factors, namely *perceived usefulness* and *perceived ease of use*. Perceived usefulness refers to the users' belief that a certain technology can deliver value. Perceived ease of use refers to the level of effort that a user would be required to

undertake to use the technology. TAM is extracted from the theory of reasoned action [42], which assumes that, to predict user behavior, those two external variables should be linked to specific intentions. These factors are viewed as indicators of an individual's willingness to use a particular technology and, consequently, impact the probability of its adoption.

The TAM is a suitable framework for creating questions aimed at determining energy stakeholders' willingness to use new technology such as AI. This is because perceived usefulness and perceived ease of use are two factors that align with common concerns that energy stakeholders may have when considering the adoption of AI.

Firstly, energy stakeholders may question the usefulness of AI in energy management because they may not fully understand how the technology works and what benefits it can provide. Additionally, they may have concerns about the reliability and accuracy of AI algorithms and models, as well as potential ethical implications and risks associated with the use of AI in decision-making processes.

In the context of energy management, stakeholders may also be more familiar with traditional methods and practices, and therefore may be skeptical about the effectiveness of AI in improving energy efficiency, reducing costs, or enhancing sustainability. Furthermore, they may have specific requirements and constraints that they believe cannot be addressed by AI technology.

To address these doubts and concerns, it is important to gather stakeholders' perceptions and expectations about the usefulness of AI in energy management, as well as to educate them about the potential benefits and limitations of the technology. Utilizing a framework such as the TAM can help in assessing stakeholders' perceptions and in identifying areas where further education and clarification may be needed. The questionnaire designed to gather stakeholders' perceptions on AI's usefulness can aid in assessing whether the technology meets the expectations and needs of the stakeholders.

Secondly, stakeholders may be apprehensive about the ease of use of AI. The TAM framework emphasizes that the technology must be perceived as easy to use to ensure its successful adoption. The questionnaire can help in assessing the stakeholders' perception of the user-friendliness of AI and simplicity of its operations. Therefore, the TAM framework is used in developing a questionnaire that can help to ensure that stakeholders' concerns and expectations are accurately captured. It can provide valuable insight into the likelihood of the successful adoption of AI technology in the given context of energy management, as illustrated in Figure 6.



Figure 6. Technology acceptance model.

The questionnaire is divided into three sections. The first one is related to the demographic characteristics of the participants, including age, gender, educational level, occupation, and employment space (public, private, or both). The aim of the second section is to define the AI literacy level (perceived usefulness) of the participants, while the third section attempts to define the perceived ease of use of the provided AI tools. In the second and third sections, both qualitative and quantitative questions were included. The quantitative questions were used as scoring components and facilitated the clustering of the specialists that participated in the survey. On the other hand, the qualitative questions were used as descriptors of the market trends and the general attitude toward AI infusion in the energy sector. In Tables 1–3, the different questions are separated in a way that reflects the described process.

 Table 1. Perceived usefulness and AI literacy level.

Question	Perceived Usefulness
Q1	Competency level in the English language
Q2	Level of literacy in AI and related expertise
Q3	Years of Experience in the sector
Q4	In the system use, were you adequately informed for every change that the system performs?
Q5	In the system use, to what extent your previous knowledge on the use of information systems has helped you understand how to use the present system?
Q6	In the system use, to what extent you needed a button for help?
Q7	To what extent were the system messages informative?
Q8	To what extent did you need to learn new functions to operate the system?
Q9	To understand the results, did you need more information that was missing from the system?
Q10	Did you need more automatic recommendations on how to use the results of the system?
Q11	Did you need more explanations on how the results of the system were generated?
Q12	Did you trust the systems results?
Q13	Did you cross-check the results using other methods?
Q14	Were the results satisfying?

Table 2. Perceived ease of use and AI usability.

Question	Perceived Ease of Use
Q1	In the system use, how easily can you predict what results are given by the functions of the system (e.g., what happens if you press a button)?
Q2	In the system use, how easily can you understand what the current situation of the system is based on your previous actions in the system?
Q3	In the system use, how easily could you predict what actions you needed to perform in similar situations within the system
Q4	In the system use, were menus and figures easy to understand in the whole of the system?
Q5	In the system use, how easy was to undo some action when needed?
Q6	In the system use, to what extent could you see the results of each of your actions?
Q7	Were the results compatible with other external sources and expertise?
Q8	Did you find the overall system useful?
Q9	Did you find the overall system easy to use?
Q10	Was the experience of using the system satisfying for you?
Q11	Would you use this system in the future?

Question	Determining Perception of AI
Q1	AI incorporation in the energy sector will result in (mark as many as appropriate)
Q2	Suggested level of explainability of a software and decision support system incorporating AI in the energy sector (mark only one)
Q3	Suggestions of potential actions towards safer, more efficient, user-friendlier, and faster incorporation of AI in the energy sector
Q4	AI will radically transform the energy sector (mark only one)
Q5	AI incorporation in the energy sector imposes threats and ethical concerns with regard to (mark as many as appropriate)
Q6	Which one of the following do you believe is the most likely cause impeding the incorporation of AI in the energy sector (mark only one)?

Table 3. Qualitative questions to define trends.

5.1. Questions Concerning AI Literacy for the Perceived Usefulness

Indeed, we use TAM to align survey outcomes with application deployment requirements to ensure higher adoption rates and smaller learning curves. For that purpose, the survey questionnaire is structured in such a way as to facilitate scoring with regard to usefulness (as related to AI) and perceived ease of use (as associated with a more generic approach to AI capabilities and tools provided by the system). By retrieving valuable input by its main potential users and stakeholders, a more concrete definition is provided of the required depth of system explainability. Recognizing users' attitude toward AI potential and their level of AI literacy, especially with regard to energy system automation and predictive capabilities of the proposed models, the developer can outline and build with intention and purpose better systems, and, at the same time, deliver tools for the circular economy.

The perceived usefulness of AI is strongly connected to AI literacy, which refers to the ability to understand, use, and critically evaluate AI technologies and their implications. AI can be considered as a new form of technological advances and thus is connected to the concept of technological literacy, which encompasses a wide range of knowledge, skills, and attitudes necessary to adapt and thrive in a rapidly changing technological environment. Technological literacy is considered an essential component of modern education and workforce preparedness by organizations such as the International Technology Education Association [43] and the National Research Council (NRC) [44]. Digital literacy, a subset of technological literacy, focuses on the effective and responsible use of digital technologies.

In turn, AI literacy is a subset of digital literacy. AI literacy is defined as a set of competencies that enable individuals to critically evaluate AI technologies, communicate and collaborate effectively with AI, and use AI as a tool online, at home, and in the workplace [45,46]. The more AI-literate a person is, the more likely they are to understand the potential benefits and applications of AI technologies, leading to a higher perception of usefulness and adoption. Literacy in a particular domain, such as digital or AI literacy, can influence an individual's perception of the ease of use and usefulness of related technologies. Energy market trends are another external variable that could impact perceived usefulness and predict technology acceptance.

Perceived usefulness is closely linked to information literacy skills because it is primarily utilized to attain specific objectives that are not naturally associated with the use of the skill itself. Davis provides the definition of perceived usefulness as the extent to which a person believes that the utilization of a particular method or technique would enhance their job performance or routine responsibility, based on the notion that the acquired capacity will improve their performance.

Research studies such as [47] suggest that pre-service teachers who have a better understanding of the potential benefits and applications of technology are more likely to perceive it as useful and thus demonstrate a higher acceptance of it. This increased acceptance could be attributed to their increased familiarity and competence in using technology as part of their education, leading to a greater understanding and self-efficacy in integrating technology into their teaching practices. Higher literacy in certain domains may result in higher self-efficacy.

Although not directly focused on technology literacy, research studies such as [48] investigate the relationship between computer self-efficacy, facilitating conditions, and perceived usefulness, which may be related to technology literacy. Additionally, the study in [49] applies an extended TAM that touches upon the relationship between an individual's background, experience, and perceived usefulness, which may relate to technology literacy.

More specifically, the following questions were created in the context of AI literacy and the perceived usefulness:

Q1: *Competency level in the English language* Competency level in the English language is often deemed necessary for using advanced computer systems, even though efforts have been made to increase accessibility in other languages; therefore, since language is the driver of expression, competency in English is an important variable for perceived usefulness and AI literacy, as lack of knowledge in the primary expressive tool in the digital domain may be considered as a factor of lower related literacy [50,51].

Q2: *Level of literacy in AI and related expertise* This question directly relates to the level of AI literacy.

Q3: Years of experience in the sector As previously mentioned, literacy in a specific technology is positively correlated with acceptance by energy stakeholders and years of experience with the technology is a contributing factor.

Q4: In the system use, were you adequately informed for every change that the system *performs*? While developing the system, efforts were made to facilitate user navigation, with consideration for the system's usability for both experts and less experienced users, which led to the inclusion of Q4 in the AI literacy evaluation, although it could also be placed in the usability and perceived ease of use category.

Q5: In the system use, to what extent your previous knowledge on the use of information systems has helped you understand how to use the present system? This considers prior knowledge as a factor in measuring literacy and self-efficacy, while it can also be seen as a variable in assessing perceived usefulness.

Q6: In the system use, to what extent you needed a button for help? One's AI literacy level can be associated with the perception of AI systems and related capabilities if they require more help, indicating that Q6, while potentially related to usability, is considered a measure of AI literacy.

Q7: To what extent were the system messages informative? As in Q4, at least some attention was given for the facilitation of the user toward navigating the system. For that reason, we considered that, for an expert, navigation should not present major issues. A beginner, however, with lesser capabilities, i.e., a lower AI literacy level, would score lower, and for that we wanted to separate it from the second group of questions, related to usability.

Q8: To what extent did you need to learn new functions to operate the system? The requirement of acquiring new knowledge should be considered part of literacy improvement or a measurement of literacy. Since AI literacy is part of perceived usefulness, Q8 should remain in this group.

Q9: To understand the results, did you need more information that was missing from the system? This question is meant to be considered in conjunction with Q4, Q5, and Q6.

Q10: *Did you need more automatic recommendations on how to use the results of the system?* A greater understanding and self-efficacy of how technology is incorporated into one's practice is an indicator of perceived usefulness and related to literacy as also analyzed in the previous sections.

Q11: *Did you need more explanations on how the results of the system were generated?* This question relates to the "Black Box" concern in the same context as Q10.

Q12: *Did you trust the systems results?* System trust is a basis of system understanding and thus the ability to understand the system should be considered as related to the users' literacy in AI and, by association, perception of usefulness.

Q13: Did you cross-check the results using other methods? In the same context as Q12.

Q14: *Were the results satisfying*? Since the system architecture and especially machine learning models have been peer-reviewed and tested by specialists, the results should be satisfying to those who understand the methods used and the potential of AI in this domain. The extent to which one could describe the methods and that potential is indeed a metric of his/her literacy and, for that reason, Q14 is included here as a variable of perceived usefulness.

In Table 1, the whole set of questions that were created to identify the AI literacy level relating to the perceived usefulness of stakeholders is summarized.

5.2. Questions about Perceived Ease of Use and Usability

Davis (1989) defined perceived ease of use as the degree to which an individual perceives that using a specific system would be effortless and hassle-free. In line with this, Zhu, Linb, and Hsu emphasized that perceived ease of use signifies the degree to which an individual accepts that using certain technology would be easy and trouble-free, and that system characteristics can influence ease of use and system usage. In [52], the authors identified several factors that may influence the ease of use of modern resources, such as computer self-efficacy, Internet self-efficacy, and information anxiety, among others [47]. Information anxiety, in particular, can affect the perceived ease of use, as access to an overwhelming amount of information is a significant challenge in today's information age. In their study, [52] extended the TAM to include factors such as experience, voluntariness of use, and objective usability. They indirectly addressed the topic of technology literacy by examining familiarity with technology. Moreover, [47] proposed a theoretical extension of the TAM based on four longitudinal field studies. In [53], the authors provide a comprehensive overview of usability engineering methods to evaluate and improve usability. Nielsen argued that designing systems with good usability leads to a higher perceived ease of use, which in turn results in more positive user experiences and increased technology adoption [54]. Ref. [55] points out that despite the ongoing research on user interface usability, the fast-paced advancements in AI bring new paradigms and quality demands, such as explainability and trustworthiness, which are often neglected in the AI literature, yet certain usability factors are crucial for human–AI interaction and for ensuring explainability and trustworthiness. This is particularly important as AI technology grows rapidly, using many kinds of algorithms and sophisticated techniques. The whole set of questions that were created to identify the usability level relating to the perceived ease of use of stakeholders relating to the eight analytics services, which were described in previous sections, is summarized in Table 2.

The group of questions from Q1 to Q6 focuses on evaluating the usability of AI tools within the system, as systems designed with good usability tend to have a higher perceived ease of use, and all questions starting with the phrase "in the system use" investigate the properties and functions of the system as implemented in the analytics services. More specifically, the reasoning underlying each question is presented below:

Q7: Were the results compatible with other external sources and expertise? This question evaluates the system's compatibility with external sources and the user's personal experience, which is an indicator of usability and perceived ease of use, as per the TAM literature, that suggests experience, voluntariness of use, and objective usability as components of perceived ease of use.

Q8: *Did you find the overall system useful?* This question is considered a variable of perceived ease of use in that it assesses the overall usefulness of the system, which includes not just the technology of AI but also its implementation and integration into the system as a whole.

Q9: *Did you find the overall system easy to use*? This question measures the ease of use of the system.

Q10: *Was the experience of using the system satisfying for you?* This question evaluates the overall satisfaction of the user's experience using the system, which contributes to the objective usability of the system.

Q11: *Would you use this system in the future?* This question assesses the likelihood of the user using the system in the future, which also contributes to the objective usability of the system.

5.3. Qualitative Questions to Define Energy Market

When using TAM to assess the acceptance of a system, it is often necessary to consider additional variables. In this study, we took market trends into account as an external factor closely related to acceptance. Evaluating product placement or more abstract concepts, such as acceptance behavior, requires considering both market trends and macroeconomics, which are interconnected factors that predict TAM. Hence, in the last section of the questionnaire, qualitative questions are created. These questions are qualitative because they ask respondents to mark their beliefs, opinions, and suggestions rather than providing numerical data. They do not involve any numerical or statistical analysis and their results cannot be quantified. Instead, they provide insights into the participants' perceptions, attitudes, and ideas related to the incorporation of AI in the energy sector and as such they are meant to define energy market trends. The whole set of qualitative questions that were created to define energy market trends is summarized in Table 3, while the full descriptive statistics are presented in the Appendix A.

More specifically, the reasoning underlying each question is presented below:

Q1: AI incorporation in the energy sector will result in (mark as many as appropriate) This question is important for identifying the potential benefits of incorporating AI into the energy sector. By understanding the potential benefits, businesses and investors can assess the market opportunity and determine where to allocate resources.

Q2: Suggested level of explainability of a software and decision support system incorporating AI in the energy sector (mark only one) The level of explainability is an important factor in the adoption of AI systems. By understanding what level of explainability is necessary for the energy sector, businesses and regulators can identify what is required for AI systems to be adopted at scale.

Q3: Suggestions of potential actions towards safer, more efficient, user-friendlier, and faster incorporation of AI in the energy sector This question is important for identifying the actions that need to be taken to ensure that AI is adopted safely and efficiently in the energy sector. By understanding the necessary actions, businesses and regulators can work to support the development and adoption of AI in the energy sector.

Q4: AI will radically transform the energy sector (mark only one) This question is important for understanding the potential impact of AI on the energy sector. By understanding the potential for transformation, businesses and investors can identify opportunities for growth and disruption in the market.

Q5: AI incorporation in the energy sector imposes threats and ethical concerns with regard to (mark as many as appropriate) This question is important for identifying the potential risks and ethical concerns associated with the adoption of AI in the energy sector. By understanding these concerns, businesses and regulators can work to mitigate risks and ensure that AI is developed and adopted responsibly.

Q6: Which one of the following do you believe is the most likely cause impeding the incorporation of AI in the energy sector (mark only one)? This question is important for identifying the barriers to the adoption of AI in the energy sector. By understanding the most significant barriers, businesses and regulators can work to address these issues and support the development and adoption of AI in the energy sector.

6. Energy Sector Survey and Analytics on Requirements of Tailored AI Explainability Levels

According to [56,57], a sample size of 20 to 40 participants is adequate for reliable and high-quality results in addressing AI-specific usability and design concerns. Therefore, a survey was conducted among 20 energy sector specialists to determine the required tailored AI explainability levels for the developed system. The participants were given an electronic questionnaire that was created based on the TAM, which was adapted to the technology of AI and HCI principles on the usability of the developed system and was presented and discussed in the previous section.

The participants' demographics were as follows: 30% were employed in a public company, 45% in a private company, and 25% both in a public and private company. Their average employment duration was 2 years with a minimum of 1 and a maximum of 5 years in this particular sector. A total of 75% of the participants identified as men and 25% as women and work or/and live in the E.U. (Figure 7). The primary language of the participants was mainly Italian (6), followed by Spanish (5). The average age of the participants was 36.6 years, with a minimum age of 22 years and a maximum age of 58 years. The highest educational level of the participants was that of Ph.D status (2 out of 20), with the majority holding a master's degree (15 out of 20) and the rest having acquired relevant certifications or equivalent to certifications degrees (3 out of 20). The electronic questionnaire was anonymously filled in, no detailed personal data were collected, and the participants submitted their consent prior to filling in the questionnaire.



Figure 7. Demographic characteristics.

6.1. Quantification Methodology

As stated in the previous section, to better define the requirements of AI explainability depth and related usability functions, a clustering analysis was performed on the questions presented in Tables 1 and 2 that were answered using a Likert scale, ranging from 1 (lowest) to 5 (highest). Due to the small amount of participants, which, however, was sufficient for the purpose of this study, silhouette coefficients were utilized to define the correct amount of clusters and the *K*-means clustering algorithm was employed to partition the group of answers into the defined number of clusters. Silhouette coefficients are values that measure

the similarity of an object to a cluster. Silhouette coefficients range from -1 to +1 and their exact value is indicative of the similarity of an object to its cluster and its dissimilarity to the other clusters [50,58]. *K*-means clustering is a quantification method of unsupervised learning that partitions observations in *K* clusters based on the mean (centroid) of each cluster and subsequently approximates each observation with that mean (centroid) [59]. Since the number of questions/observations is more than two and it would be difficult to define the clusters in the multidimensional space, the PCA reduction technique was used for simplification purposes. Principal component analysis (PCA) converts the full features into fewer features based on their contribution in explaining a component (variance explained) [60].

6.2. AI Literacy Score, Clusters, and Perceived Usefulness

Using the proposed methodology of K-means clustering, the silhouette coefficients used to define the number of clusters, and the PCA technique used to visualize those clusters, the answers of the questions of Table 1 have been analyzed. The results can be seen in Figure 8. According to the analysis, there are two different clusters related to the corresponding participants' AI literacy level, as revealed by the silhouette coefficient and seen in Figure 8, in which the highest value corresponds to the number of clusters. The primary characteristics, based on the average score of each answer, can be seen in Table 4. To plot the different clusters, we have used three components that, as can be seen in the principal component analysis of Figure 8, explain more that 58% of the total variables (14). The average score of the first cluster is 3.45, while that of the second cluster is 2.93. There is greater separation (above 0.70 decimal points) between the two clusters in Questions 1, 3, 9, 13, and 14. Question 1 is related to the users' competency in the English language. Question 3 is related to the years of working experience in the sector. Question 9 corresponds to the required explainability of the D.S.S. result and the need of additional information that is perceived as missing. Question 13 measures the frequency that the user validated results using external sources and finally Question 14 is related to the degree of user satisfaction with the results provided by the system.



Figure 8. Defining the layers of explainability.

To summarize, two levels of explainability seem necessary, corresponding to the characteristics of the two different users. Thus, at least years of working experience, level

of understanding of the English language, and potentially offering a localized environment based on user language requirements should be taken into consideration. Trialability is also an important factor as described in Questions 13 and 14, where related literature should be additionally provided, which users should be encouraged to study. In both clusters, AI literacy is either average or below average (Q2), and for that reason the usefulness of the different explainability methodologies (local–global XAI) should be provided and detailed. The experienced users felt that system messages were adequate, but the second less experienced group of users felt that some actions performed by the provided AI pipeline were not sufficiently explained.

Question	Cluster 1 (Mean)	Cluster 2 (Mean)
Q1	4.25	3.5
Q2	2.5	2.0833
Q3	2.875	1.4166
Q4	4.000	2.8333
Q5	2.750	2.7500
Q6	3.875	3.0833
Q7	3.000	3.000
Q8	3.750	3.166
Q9	3.750	3.2500
Q10	3.125	3.4166
Q11	3.000	3.166
Q12	3.625	3.416
Q13	3.875	2.666
Q14	4.000	3.333
Total Count	8	12
Average	3.45	2.93

Table 4. Perceived usefulness and AI literacy level—Average Scores.

6.2.1. AI Tools Usability Score, Clusters, and Perceived Ease of Use

In this section, we assess the perceived ease of use of the AI tools offered. Similarly to the previous section, a segmentation methodology was used to define the optimum number of clusters. Specifically, five clusters were identified, using the silhouette coefficients, where five is indicated as the highest value of a total of six cluster scenarios tested. Using PCA, we reduced the plotted space from 11 components in total in the questions to 3 components, i.e., we reduced the feature space down to a three-dimensional one. The components explain almost 60% of the variance and the corresponding *K*-means plotted clusters can be seen in Figure 9. The five different clusters assume different scores, which indicates different abilities and different perceptions as per the usability of AI tools offered by our system (Table 5). Cluster 3, within which most of the users are gathered, has a slightly above average perception of 2.98 for the perceived ease of use of the system AI. Clusters 2, 4, and 5, which hold the highest average scores and also include 35% of the total participants, score above 3.5, which should be considered a strong indicator of a general acceptance of the offered tools, with room, though, for improvements.

The questions with the lowest scores are Question 3 in Cluster 1, Question 1 in Cluster 4, and Question 4 in Cluster 5. These questions are stated as follows:

- In the system use, how easily can you predict what results are given by the functions of the system (e.g. what happens if you press a button)?
- In the system use, how easily could you predict what actions you needed to perform in similar situations within the system
- In the system use, were menus and figures easy to understand in the whole of the system? The above mentioned questions indicate a difficulty in understanding and navigating

through the exported results of the AI pipeline. Again, here as in the previous section, an explainability framework that considers different levels of AI-related capabilities can indeed offer a greater understanding of the system underlying value and thus increase adoption via improving the perception of the usability of AI and the general usefulness of extracted recommendations and automated suggestions related to the different energy analytical services provided by the ML pipeline.

It is also important to note that the general sentiment, as reflected by Question 8 on the overall system usefulness, scores the highest for all clusters, indicating that the needs of the energy sector are well served and identified.



Figure 9. Defining the layers of perceived ease of use and AI usability.

Question	Cluster 1 (Mean)	Cluster 2 (Mean)	Cluster 3 (Mean)	Cluster 4 (Mean)	Cluster 5 (Mean)		
Q1	2.75	3.75	2.88	2.0	2.5		
Q2	2.00	3.75	3.00	5.0	3.0		
Q3	1.75	2.75	3.00	4.0	3.0		
Q4	2.00	3.00	3.22	5.0	1.0		
Q5	2.25	3.00	2.44	3.0	3.5		
Q7	3.50	4.00	3.00	3.0	4.0		
Q8	3.75	4.50	3.33	5.0	5.0		
Q9	3.50	3.50	2.88	4.0	4.0		
Q10	3.50	3.75	3.11	3.0	5.0		
Q11	4.25	3.50	3.33	5.0	5.0		
TotalCount	4	4	9	1	2		
Average	2.93	3.59	2.98	3.63	3.5		

Table 5. Perceived ease of use and AI usability—Average scores.

6.2.2. Market Trends, towards an AI-Driven Energy Sector

The answers provided in the last portion of the survey can be seen in Figure 10. The main concern of the participants, with regard to AI incorporation in the energy sector, is related to data privacy (80%) and security. The general consensus is that AI will further decentralize the sector (80%) and will facilitate faster decarbonization (65%) and increase energy savings (65%). From the following questions, which are also highly suggestive of the previous results analyzed and also validate the clustering hypothesis, it appears that

the most likely cause that would impede AI incorporation is, by a majority of 60%, the lack of explainability (30%) and lack of required training of users to the available software (30%). The suggested level of explainability should be at a level of limited detail for non-technical users (45%), followed by a level that corresponds to deeper technical detail, suitable for more technical users.

The vast majority (45%) expect AI to transform the energy sector in the 2030s, while 30% see this transformation taking place in the 2020s. Moreover, 10% of the participants express the opinion that AI will never transform the energy sector and 15% that the transformation will occur beyond the 2030s. Again, the results signify two different kind of users in the participant pool.



Figure 10. Qualitative-based questions.

7. Incorporation and Implementation of Tailored XAI in the DSS

According to TAM, analyzed in the previous sections, the perceived ease of use (AI usability) and perception of usefulness (AI literacy level) are communicating vessels, interconnected in the users' consciousness. We have identified two different clusters in our data, related to their analytical experience, and have quantified their AI literacy level. More clusters have been proposed based on the perceived ease of use, where some key points were identified and should be addressed via design modifications. In Figure 11, the generalization of our methodology can be seen. The different XAI tools are provided depending on a defined personalization.

For each cluster, different tools are proposed with some tools overlapping with regard to AI explainability. For Cluster 1, a technical analysis of the ML models should be provided, using local and global XAI, bias analysis, and model architecture characteristics. The related literature should also be provided. Cluster 2, which consists of less experienced users, should have access to basic textual description of the process. A semantic grouping of inputs and output [61] combined with cost benefit analysis and contribution to the circular economy via AI applicability in the sector should provide incentives and build trust while simultaneously overcoming validation and regulation challenges.



Figure 11. Grouping of XAI tools and inner clusters.

To address the problem of perceived usefulness and perceived ease of use, following the proposed methodology outlined in Figure 11, we offer a preview of screens used to explain the predictive technology used for one of the developed applications, incorporating the presented technologies.

7.1. Model Architecture and Benchmarks

In this subsection, we present a concrete example of a practical application of our proposed tailored XAI methodology. Specifically, we apply our methodology with regard to analytic service AS04 (Figure 2). From Figures 2, 4 and 5, it is clear that AS04 is to be used by stakeholders BUI (consumers, customers, residents of buildings, occupants), FMB (facility managers, building operators), and INV (project developers, investing funds). These stakeholders have very different backgrounds, preferences, skills, and abilities, and thus different perceptions of ease of use and usefulness of service AS04. Those differences require different explanation approaches to ensure greater and faster adoption rates. It should be mentioned though that even though stakeholders are in defined groups, within the groups, people with different abilities and skills exist. For that, we employ TAM to have a better understanding of those abilities and provide accurate personalization.

In Figure 12, the model architecture of AS04 is presented. An ensemble of key models is used to make predictions based on a probability of a three-class outcome, using as feature inputs (1) the energy consumption before renovation, (2) the cost of renovation, (3) the planned CO_2 reduction as detailed by the constructor specifications, (4) the building age in years, and (5) the total heating area defined by the building envelope [40].

Next, the overall model outputs the probabilities for each class. Based on the investment potential, which is the relationship between energy consumption reduction and cost of investment, the model considers three classes. For Class A, the potential for investment is optimum, while for Class B it is medium and the project should be only partially financed. For Class C, the project should not be financed. The different models are stacked and use linear regression as the activation function to make predictions.



Figure 12. Model architecture—benchmarks, change benchmark figures.

Using an ensemble of models, an average prediction is provided, ensuring optimized results for a given problem using different algorithms for a better definition and boosted solution overall. In the benchmark section, the accuracy of each model is calculated, alongside the final result of the stacked model. Finally, the correlation between energy consumption and investment cost for each class is shown. In this section, we look into the proposed methods outlined in Figure 12, where the model architecture for Cluster 1 and benchmarks for Cluster 2 are touched upon.

7.2. Textual Descriptors—Semantic Grouping

Textual descriptions are comments for users to understand the reasoning behind the development of an application and the personal and macro-economic benefits of using it. Some key points that have been analyzed in this paper are used as descriptors of usefulness, such as the idea of circular economics and the semantic grouping of applications and stakeholders. The relation between inputs and outputs [61] is also an important factor that can increase adoption, add trust to the system, and clarify the system results. In Figure 13, the semantic relation of inputs and outputs is presented [40]. In the middle section, the data structure of each input is introduced. Each element has more information attached to it for the user to explore, thus offering an in-depth look at how the system analyzes data and outputs predictions.



Figure 13. Textual descriptors and semantic grouping.

7.3. Local XAI

In Figure 14, selected graphs show how the different features contribute to each decision for a specific observation from the analyzed dataset. We have used Shapley values to determine the different outcomes and the SHAP Python libraries to create an explainability framework for the examined case [62]. According to the observation analyzed, the true actual value refers to Class B or Class 1. The predicted array, titled as predicted in Figure 14, shows the probabilities for each class as they are extracted by the stacked model. For Class 0, the probabilities are equal to 0.16 (0.0155), while for Class 1 they are equal to 0.924 and for Class 2 they are equal to 0.06 (0.0597). Our model has correctly predicted the class as 1.

To see how the different features contributed to the prediction, we have used SHAP force and waterfall plots. In the force plot, the different interactions between the features and the decision per class is outlined, arriving at the predicted solution probability of outcome, symbolized as f(x). On the other hand, the waterfall plots similarly show the different responses of each parameter. At the bottom of the graph, the expected output is shown, while the top of the graph indicates how the different features contributed to the actual prediction extracted by the model. Since Class 1 was indeed the correct output, in both graphs, all features in Class 1 point to the right result and the proposed investment should be only partially funded.



Figure 14. Local XAI.

7.4. Global XAI

In Figure 15, selected graphs show how the different features contribute to the final prediction for each class. Again, Shapley values have been utilized, alongside the SHAP Python libraries for explaining the results visually. Feature importance is first analyzed for each class for all inputs. Cost is the highest contributing factor in all cases, but the contribution of other factors varies for the different classes.



Figure 15. Global XAI.

The dependency plots are then utilized to summarize the dependence between the actual cost and the actual contribution of the cost to the decision eventually made by the model, labeled as SHAP value for cost on the y-axis of each scatter graph. The SHAP value for the examined variable shows the extent to which knowing this particular variable affects the prediction of the model. On the opposite side of the y-axis, a feature that is closely related to the examined variable is tracked and the extent of the effect to the examined variable is differentiated by color.

7.5. User Interactions and Provide Outcomes

For the global and local XAI, where, in the provided examples, Shapley values were extracted, Algorithms A4 and A5 were utilized (see Appendix A). Via an interactive environment and depending on the personalization options, the user can choose any datapoint used in the training process of the neural networks (NNs) utilized for the analytical services. Datapoints are embedded objects of the NNs and thus can be called upon and examined. Examples of this process were provided in the previous subsections. For each datapoint, an explanation is always returned and errors logged during the testing are not associated with missing datapoints or the inability of the system to compute.

8. Summary, Conclusions, and Future Work

In conclusion, this paper proposes a novel methodology for designing a tailored explainability solution for AI analytics services in the energy management sector. The methodology is based on multiple processes that take into account the specific needs and preferences of stakeholders, as well as established frameworks such as the TAM and HCI principles. The methodology is comprehensive and involves identifying key stakeholders and AI analytics services, creating a tailored explainability requirements questionnaire, conducting a survey among energy management stakeholders, performing a user-based clustering analysis to determine the required depth of explainability, and implementing the tailored XAI solution at both the local and global levels. The contribution of this methodology lies in its ability to develop an effective solution that meets the diverse needs of stakeholders in the energy management sector. The methodology is tested on a stacked neural network for the AS04 analytic service, which estimates energy savings from renovations, and aims to increase adoption rates and benefit the circular economy. The significance of AI explainability is emphasized as an ethical approach and a tool to build user trust and ensure faster adoption rates, especially in the energy sector, where AI can provide a more sustainable future.

In summary, the contributions of this research are the following: firstly, the agile development of an IDSS considers XAI as a main feature and ensures that it is tailored to specific user needs. Secondly, to facilitate that personalization, TAM was used to cluster stakeholders based on the perception of usefulness and ease of use. Perception of usefulness was associated with AI literacy and perceived ease of use with usability. These different perceptions require different explanation approaches to ensure greater and faster adoption rates and, for that, XAI tools were grouped accordingly. The limitation to be addressed in our future work is the evaluation of our methodology with real-life data as per adoption, user satisfaction, and an increase in **application utilization**. Future work can also extend the methodology to other AI applications in the energy sector and other industries, evaluate it on various metrics such as efficiency, effectiveness, scalability, and interpretability, provide more in-depth explanations to users, and examine the impact of explainability on the adoption rate of AI systems in other industries. This can help build more responsible and safe spaces for the deployment of AI systems.

Author Contributions: Conceptualization, D.P.P. and M.V.; methodology, V.M., M.V., G.A.T., and H.D.; software, D.P.P. and E.S.; validation, V.M., G.A.T., and H.D.; data curation, D.P.P. and E.S.; writing—original draft, D.P.P., E.S., M.V. and G.A.T.; writing—review and editing, G.A.T.; visualization, D.P.P.; supervision, V.M., M.V. and H.D.; project administration, V.M. and H.D. All authors have read and agreed to the published version of the manuscript.

Funding: The work presented is based on research conducted within the framework of the H2020 European Commission project BD4NRG (Grant Agreement No. 872613).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Regarding the questionnaires, the anonymous collection of data was purely for research purposes via Google forms. The data were collected by the Decision Support Systems Laboratory School of Electrical and Computer Engineering.

Data Availability Statement: Anonymous data collected via electronic questionnaires. Further data details included in Appendix A.

Acknowledgments: The content of the paper is the sole responsibility of its authors and does not necessary reflect the views of the EC.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

ML	Machine Learning
AI	Artificial Intelligence
XAI	Explainable Artificial Intelligence
RAI	Responsible Artificial Intelligence
PDPs	Partial Dependence Plots
ALEs	Accumulated Local Effect Plots
LIMEs	Local Interpretable Model-agnostic Explanations
SHAPs	SHapley Additive exPlanations
TAM	Technology Acceptance Model
TOFHLA	Test of Functional Health Literacy in Adults
ICE	Individual Conditional Expectation
HCI	Human–Computer Interaction
DSS	Decision Support System
IDSS	Intelligent Decision Support System
RESs	Renewable Energy Sources
NNs	Neural networks

Appendix A

Table A1. Descriptive statistics of survey.

	Count	Mean	STD	Min	25%	50%	75%	Max
Age	20.0	36.6	10.31	22.0	29.5	35.5	42.0	58.0
Children	20.0	0.85	0.988	0.0	0.0	0.5	2.0	3.0
Competency level in the English language	20.0	3.8	0.833	2.0	3.0	4.0	4.0	5.0
Level of literacy in AI and related expertise	20.0	2.25	0.91	1.0	2.0	2.0	3.0	4.0
Years of Experience in the sector	20.0	2.0	1.33	1.0	1.0	1.5	2.25	5.0
"In the system use, how easily can you predict what results are given by the functions of the system (e.g., what happens if you press a button)?"	20.0	2.95	0.75	2.0	2.0	3.0	3.25	4.0
"In the system use, how easily can you understand what the current situation of the system is based on your previous actions in the system?"	20.0	3.05	0.88	1.0	3.0	3.0	3.25	5.0
"In the system use, were you adequately informed for every change that the system performs?"	20.0	3.3	0.97	1.0	3.0	3.0	4.0	5.0
"In the system use, to what extent your previous knowledge on the use of information systems has helped you understand how to use the present system? "	20.0	2.75	0.966	1.0	2.0	3.0	3.25	4.0
"In the system use, how easily could you predict what actions you needed to perform in similar situations within the system?"	20.0	2.75	0.71	1.0	2.0	3.0	3.0	4.0
"In the system use, were menus and figures easy to understand in the whole of the system?"	20.0	2.8	1.10	1.0	2.0	3.0	3.0	5.0
"In the system use, to what extent you needed a button for help?"	20.0	3.4	0.82	1.0	3.0	3.0	4.0	5.0
"In the system use, how easy was to undo some action when needed?"	20.0	2.65	0.87	1.0	2.0	3.0	3.0	4.0
"In the system use, to what extent could you see the results of each of your actions?"	20.0	2.9	0.91	1.0	2.0	3.0	3.0	5.0
"To what extent were the system messages informative?"	20.0	3.0	0.85	1.0	3.0	3.0	3.0	5.0
"To what extent did you need to learn new functions to operate the system?"	20.0	3.4	0.753	2.0	3.0	3.0	4.0	5.0
"To understand the results, did you need more information that was missing from the system?"	20.0	3.45	0.60480	3.0	3.0	3.0	4.0	5.0
"Did you need more automatic recommendations on how to use the results of the system?"	20.0	3.3	0.73	2.0	3.0	3.0	4.0	5.0
"Did you need more explanations on how the results of the system were generated?"	20.0	3.1	0.78	2.0	3.0	3.0	3.25	5.0
"Did you trust the systems results?"	20.0	3.5	0.82	1.0	3.0	4.0	4.0	5.0
"Did you cross-check the results using other methods?"	20.0	3.15	1.13	1.0	2.0	3.0	4.0	5.0
"Were the results satisfying?"	20.0	3.6	0.68	2.0	3.0	4.0	4.0	5.0
"Were the results compatible with other external sources and expertise?"	20.0	3.4	0.75	2.0	3.0	3.0	4.0	5.0
"Did you find the overall system useful?"	20.0	3.9	0.788	3.0	3.0	4.0	4.25	5.0
"Did you find the overall system easy to use?"	20.0	3.3	0.65	2.0	3.0	3.0	4.0	5.0
"Was the experience of using the system satisfying for you?"	20.0	3.5	0.76	2.0	3.0	3.0	4.0	5.0
"Would you use this system in the future?"	20.0	3.8	0.83	2.0	3.0	4.0	4.0	5.0

Algorithm A1 Feature Engineering and analysis.

1: Initial:

- 2: Load data
- 3: Print the descriptive statistics of the loaded data.
- 4: Save statistics to JSON file.
- 5: GetsNumericNonNumerics:
- 6: Load data
- 7: Set the path for file operations.
- 8: Count the number of columns in the data.
- 9: Identify and save non-numeric columns from the data to JSON file.
- 10: Identify and save numeric columns from the data to JSON file.
- 11: **PrimaryAnalysis:**
- 12: Load data as df
- 13: Set the path for file operations.
- 14: Count the number of columns in the data.
- 15: **for** each column in **df do**
- 16: Determine the datatype of the column.
- 17: **if** datatype is numeric **then**
- 18: Print "New df" to signal data analysis.
- 19: Analyze and Plot "New df".
- 20: end if
- 21: if datatype is non-numeric then
- 22: Print "New df2" to signal data analysis.
- 23: Analyze and Plot "New df2".
- 24: end if
- 25: **end for**
- 26: Save "New df" to JSON file.
- 27: Save "New df2" to JSON file.
- 28: End of Algorithm.

Algorithm A2 TAM Algorithm.

1: Condition:

- 2: Load data "New df"
- 3: Set path for the JSON file and the output directory.
- 4: Define a list of wanted column names based on Condition.
- 5: Load the desired dataset from a JSON file using the defined list.
- 6: Copy the specific columns from the loaded data to a new dataframe.
- 7: Perform hierarchical clustering on the new dataframe
- 8: Save Plots To File
- 9: Save paths and labels to JSON file.
- 10: End of Algorithm.

Algorithm A3 ModelDeconstruction.

Require: Model Object (model_object), JSON Path (json_path), JSON file name (name) **Ensure:** Model Summary as a JSON file and Model Architecture as an image file

- 1: Begin
- 2: Load the model from the model_object
- 3: Generate model summary and store it into a list
- 4: Create a table using every alternate line from the list (excluding first and last two lines)
- 5: **for** every entry in the table **do**
- 6: Split the entry based on multiple spaces and remove the last part of each entry
- 7: Append the cleaned up entry to a new_table
- 8: end for
- 9: Convert new_table into a data frame (df) where the first row becomes column names
- Generate an image of the model architecture and save it as "modelArchitecture.png" at json_path
- 11: Save the dictionary (list) into a JSON file at json_path with the name specified
- 12: End

Algorithm A4 LocalExplainer.

Require: path, X_train, X_test, WhatToTest, y_test, profile, inputsList, categoryName, jsonpath

- Ensure: Several plots saved as images and a JSON file saved at jsonpath
- 1: Begin
- 2: Call LocalJson function with parameters categoryName, profile, WhatToTest, and inputsList to prepare a list
- 3: Call localModelEvaluator function with 'path' as an argument to load the model
- 4: Initialize an Explainer using the loaded model and the X_train data
- 5: Generate explanation values using the Explainer on X_test
- 6: Make a prediction using the loaded model on X_test for WhatToTest
- 7: Update "Prediction" and "Actual value" in the list with the predicted and actual values
- 8: Generate various plots using explanation values and save them as images at specified jsonpath
- 9: Update corresponding image paths in the list under "XaiImages"
- 10: Save the updated list to a JSON file at jsonpath with categoryName as the file name

Algorithm A5 GlobalExplainer.

Require: path, X_train, X_test, profile, categoryName, jsonpath

Ensure: Plots saved as images and a JSON file saved at jsonpath

- 1: Begin
- 2: Call GlobalJson function with categoryName to prepare a list
- 3: Call GlobalModelEvaluator function with 'path' as an argument to load the model
- 4: Initialize an Explainer using the loaded model and the X_train data
- 5: Generate explanation values using the Explainer on X_test
- 6: Generate a summary plot using explanation values and save it as an image at specified jsonpath. Update corresponding image path in the list under "OtherPlots"
- 7: for each feature in the profile do
- 8: Generate a dependence plot using the explanation values and feature, save the plot as an image at specified jsonpath, update the corresponding image path in the list under "DependencyPlots"
- 9: end for
- 10: Save the updated list to a JSON file at jsonpath with categoryName as the file name
- 11: End

^{11:} End

References

- 1. United Nations . Paris agreement. In Proceedings of the Report of the Conference of the Parties to the United Nations Framework Convention on Climate Change (21st Session, 2015: Paris), Paris, France, 30 November–13 December 2015; Volume 4, pp. 1–42.
- Tollefson, J.; Weiss, K.R. Nations adopt historic global climate accord: Agreement commits world to holding warming' well below' 2 [degrees] C. *Nature* 2015, 582, 315–317. [CrossRef] [PubMed]
- Doukas, H.; Nikas, A.; González-Eguino, M.; Arto, I.; Anger-Kraavi, A. From integrated to integrative: Delivering on the Paris Agreement. Sustainability 2018, 10, 2299. [CrossRef]
- 4. Carley, S.; Konisky, D.M. The justice and equity implications of the clean energy transition. *Nat. Energy* **2020**, *5*, 569–577. [CrossRef]
- Bataille, C.; Åhman, M.; Neuhoff, K.; Nilsson, L.J.; Fischedick, M.; Lechtenböhmer, S.; Solano-Rodriquez, B.; Denis-Ryan, A.; Stiebert, S.; Waisman, H.; et al. A review of technology and policy deep decarbonization pathway options for making energy-intensive industry production consistent with the Paris Agreement. J. Clean. Prod. 2018, 187, 960–973. [CrossRef]
- 6. Papadis, E.; Tsatsaronis, G. Challenges in the decarbonization of the energy sector. *Energy* **2020**, *205*, 118025. [CrossRef]
- Hope, A.; Roberts, T.; Walker, I. Consumer engagement in low-carbon home energy in the United Kingdom: Implications for future energy system decentralization. *Energy Res. Soc. Sci.* 2018, 44, 362–370. [CrossRef]
- Baidya, S.; Potdar, V.; Ray, P.P.; Nandi, C. Reviewing the opportunities, challenges, and future directions for the digitalization of energy. *Energy Res. Soc. Sci.* 2021, *81*, 102243. [CrossRef]
- Zafar, R.; Mahmood, A.; Razzaq, S.; Ali, W.; Naeem, U.; Shehzad, K. Prosumer based energy management and sharing in smart grid. *Renew. Sustain. Energy Rev.* 2018, 82, 1675–1684. [CrossRef]
- 10. Manfren, M.; Nastasi, B.; Tronchin, L.; Groppi, D.; Garcia, D.A. Techno-economic analysis and energy modelling as a key enablers for smart energy services and technologies in buildings. *Renew. Sustain. Energy Rev.* **2021**, *150*, 111490. [CrossRef]
- 11. Weigel, P.; Fischedick, M. Review and categorization of digital applications in the energy sector. *Appl. Sci.* **2019**, *9*, 5350. [CrossRef]
- Marinakis, V.; Doukas, H.; Tsapelas, J.; Mouzakitis, S.; Sicilia, Á.; Madrazo, L.; Sgouridis, S. From big data to smart energy services: An application for intelligent energy management. *Future Gener. Comput. Syst.* 2020, 110, 572–586. [CrossRef]
- Sarmas, E.; Dimitropoulos, N.; Strompolas, S.; Mylona, Z.; Marinakis, V.; Giannadakis, A.; Romaios, A.; Doukas, H. A web-based building automation and control service. In Proceedings of the 2022 13th International Conference on Information, Intelligence, Systems & Applications (IISA), Corfu, Greece, 18–20 July 2022; pp. 1–6.
- 14. Esmat, A.; de Vos, M.; Ghiassi-Farrokhfal, Y.; Palensky, P.; Epema, D. A novel decentralized platform for peer-to-peer energy trading market with blockchain technology. *Appl. Energy* **2021**, *282*, 116123. [CrossRef]
- 15. Fell, M.J. Energy services: A conceptual review. Energy Res. Soc. Sci. 2017, 27, 129–140. [CrossRef]
- 16. Gailhofer, P.; Herold, A.; Schemmel, J.P.; Scherf, C.S.; de Stebelski, C.U.; Köhler, A.R.; Braungardt, S. *The Role of Artificial Intelligence in the European Green Deal*; European Parliament: Luxembourg, Belgium, 2021.
- Arrieta, A.B.; Díaz-Rodríguez, N.; Del Ser, J.; Bennetot, A.; Tabik, S.; Barbado, A.; Garcia, S.; Gil-Lopez, S.; Molina, D.; Benjamins, R.; et al. Explainable Artificial Intelligence (XAI): Concepts, taxonomies, opportunities and challenges toward responsible AI. *Inf. Fusion* 2020, *58*, 82–115. [CrossRef]
- Meske, C.; Bunde, E.; Schneider, J.; Gersch, M. Explainable Artificial Intelligence: Objectives, Stakeholders, and Future Research Opportunities. *Inf. Syst. Manag.* 2022, 39, 53–63. [CrossRef]
- Hacker, P.; Passoth, J.H. Varieties of AI explanations under the law. From the GDPR to the AIA, and beyond. In Proceedings of the xxAI-Beyond Explainable AI: International Workshop, Held in Conjunction with ICML 2020, Revised and Extended Papers, Vienna, Austria, 18 July 2020; Springer: Berlin/Heidelberg, Germany, 2022; pp. 343–373.
- Ngo, T.; Kunkel, J.; Ziegler, J. Exploring mental models for transparent and controllable recommender systems: A qualitative study. In Proceedings of the 28th ACM Conference on User Modeling, Adaptation and Personalization, Genoa, Italy, 12–18 July 2020; pp. 183–191.
- Tsai, C.H.; Carroll, J.M. Logic and pragmatics in AI explanation. In Proceedings of the xxAI-Beyond Explainable AI: International Workshop, Held in Conjunction with ICML 2020, Revised and Extended Papers, Vienna, Austria, 18 July 2020; Springer: Berlin/Heidelberg, Germany, 2022; pp. 387–396.
- 22. Kinger, S.; Kulkarni, V. Explainable ai for deep learning based disease detection. In Proceedings of the 2021 13th International Conference on Contemporary Computing (IC3-2021), Noida, India, 5–7 August 2021; pp. 209–216.
- European Council. Fit for 55: The EU's Plan for a Green Transition. 2020. Available online: https://www.consilium.europa.eu/ en/policies/green-deal/fit-for-55-the-eu-plan-for-a-green-transition/ (accessed on 1 January 2020).
- IEA. Renewables 2021: Analysis and Forecasts to 2026. 2021. Available online: https://iea.blob.core.windows.net/assets/5ae322 53-7409-4f9a-a91d-1493ffb9777a/Renewables2021-Analysisandforecastto2026.pdf (accessed on 1 January 2020).
- Bradley, P.; Leach, M.; Torriti, J. A review of the costs and benefits of demand response for electricity in the UK. *Energy Policy* 2013, 52, 312–327. [CrossRef]
- 26. Gellings, C.W.; Smith, W.M. Integrating demand-side management into utility planning. Proc. IEEE 1989, 77, 908–918. [CrossRef]
- 27. Raza, M.Q.; Khosravi, A. A review on artificial intelligence based load demand forecasting techniques for smart grid and buildings. *Renew. Sustain. Energy Rev.* 2015, *50*, 1352–1372. [CrossRef]

- 28. Xiong, R.; Cao, J.; Yu, Q. Reinforcement learning-based real-time power management for hybrid energy storage system in the plug-in hybrid electric vehicle. *Appl. Energy* **2018**, *211*, 538–548. [CrossRef]
- 29. Duan, J.; Yi, Z.; Shi, D.; Lin, C.; Lu, X.; Wang, Z. Reinforcement-learning-based optimal control of hybrid energy storage systems in hybrid AC–DC microgrids. *IEEE Trans. Ind. Inform.* **2019**, *15*, 5355–5364. [CrossRef]
- Zsembinszki, G.; Fernández, C.; Vérez, D.; Cabeza, L.F. Deep learning optimal control for a complex hybrid energy storage system. *Buildings* 2021, 11, 194. [CrossRef]
- 31. Stefanakis, A.; Nikolaou, I. Circular Economy and Sustainability: Volume 1: Management and Policy; Elsevier: Amsterdam, The Netherlands, 2021.
- Geissdoerfer, M.; Savaget, P.; Bocken, N.M.; Hultink, E.J. The Circular Economy—A new sustainability paradigm? *J. Clean. Prod.* 2017, 143, 757–768. [CrossRef]
- Nikolaou, I.E.; Tsagarakis, K.P. An introduction to circular economy and sustainability: Some existing lessons and future directions. Sustain. Prod. Consum. 2021, 28, 600–609. [CrossRef]
- PricewaterhouseCoopers. The Circular Economy Model. 2021. Available online: https://www.pwc.com/gr/en/advisory/riskassurance/sustainability-climate-change/circular-economy-model.html (accessed on 1 January 2020).
- The European Parliament and the Council of the European Union. Directive 2009/125/EC of the European Parliament and of the Council of 21 October 2009, establishing a framework for the setting of ecodesign requirements for energyrelated products (recast). Off. J. Eur. Communities 2009, 20, 1–26.
- 36. Larman, C.; Basili, V.R. Iterative and incremental developments. A brief history. Computer 2003, 36, 47–56. [CrossRef]
- 37. Larman, C. Scaling Lean & Agile Development: Thinking and Organizational Tools for Large-Scale Scrum; Pearson Education: Bengaluru, India, 2008.
- 38. Schwaber, K. Agile Project Management with Scrum; Microsoft Press: Redmond, WA, USA, 2004.
- Sarmas, E.; Dimitropoulos, N.; Marinakis, V.; Zucika, A.; Doukas, H. Monitoring the impact of energy conservation measures with artificial neural networks. In Proceedings of the in ECEEE Summer Study, Hyeres, France, 6–10 June 2022.
- 40. Sarmas, E.; Spiliotis, E.; Dimitropoulos, N.; Marinakis, V.; Doukas, H. Estimating the Energy Savings of Energy Efficiency Actions with Ensemble Machine Learning Models. *Appl. Sci.* **2023**, *13*, 2749. [CrossRef]
- 41. Tsolkas, C.; Spiliotis, E.; Sarmas, E.; Marinakis, V.; Doukas, H. Dynamic energy management with thermal comfort forecasting. *Build. Environ.* **2023**, 237, 110341. [CrossRef]
- 42. Ajzen, I. The theory of planned behavior. Organ. Behav. Hum. Decis. Process. 1991, 50, 179-211. [CrossRef]
- 43. Dugger, W.E., Jr. Standards for Technological Literacy. Content for the Study of Technology. Technol. Teach. 2000, 59, 8–13.
- 44. National Research Council. National Science Education Standards; National Academies Press: Cambridge, MA, USA, 1996.
- 45. Lankshear, C.; Knobel, M. Researching new literacies: Web 2.0 practices and insider perspectives. *E-Learn. Digit. Media* 2007, 4, 224–240. [CrossRef]
- 46. Bughin, J.; Hazan, E.; Ramaswamy, S.; Chui, M.; Allas, T.; Dahlstrom, P.; Henke, N.; Trench, M. Artificial Intelligence: The Next Digital Frontier? Discussion Paper; McKinsey and Company: Mumbai, India, 2017.
- Venkatesh, V.; Davis, F.D. A theoretical extension of the technology acceptance model: Four longitudinal field studies. *Manag. Sci.* 2000, 46, 186–204. [CrossRef]
- 48. Teo, T. Factors influencing teachers' intention to use technology: Model development and test. *Comput. Educ.* **2011**, *57*, 2432–2440. [CrossRef]
- Hsu, C.L.; Lu, H.P. Why do people play on-line games? An extended TAM with social influences and flow experience. *Inf. Manag.* 2004, 41, 853–868. [CrossRef]
- 50. Zhou, H.B.; Gao, J.T. Automatic method for determining cluster number based on silhouette coefficient. In *Proceedings of the Advanced Materials Research*; Trans Tech Publications: Stafa-Zurich, Switzerland, 2014; Volume 951, pp. 227–230.
- 51. Hafner, C.A. Digital literacies for English language learners. In *Second Handbook of English Language Teaching*; Springer: Berlin/Heidelberg, Germany, 2019; pp. 899–918.
- 52. Durodolu, O. Technology Acceptance Model as a predictor of using information system'to acquire information literacy skills. *Libr. Philos. Pract.* **2016**, 1450, 1–28.
- 53. Nielsen, J. Usability Engineering; Morgan Kaufmann: Burlington, MA, USA, 1994.
- 54. Venkatesh, V.; Bala, H. Technology acceptance model 3 and a research agenda on interventions. *Decis. Sci.* 2008, 39, 273–315. [CrossRef]
- 55. Virvou, M. Artificial Intelligence and User Experience in reciprocity: Contributions and state of the art. *Intell. Decis. Technol.* **2023**, 17, 1–53. [CrossRef]
- 56. Alroobaea, R.; Mayhew, P.J. How many participants are really enough for usability studies? In Proceedings of the 2014 Science and Information Conference, London, UK, 27–29 August 2014; pp. 48–56.
- 57. Sauro, J.; Lewis, J.R. *Quantifying the User Experience: Practical Statistics for User Research;* Morgan Kaufmann: Burlington, MA, USA, 2016.
- Aranganayagi, S.; Thangavel, K. Clustering categorical data using silhouette coefficient as a relocating measure. In Proceedings of the International Conference on Computational Intelligence and Multimedia Applications (ICCIMA 2007), Tamil Nadu, India, 13–15 December 2007; Volume 2, pp. 13–17. [CrossRef]
- 59. Likas, A.; Vlassis, N.; Verbeek, J.J. The global k-means clustering algorithm. Pattern Recognit. 2003, 36, 451–461. [CrossRef]

- 60. Shlens, J. A tutorial on principal component analysis. *arXiv* **2014**, arXiv:1404.1100.
- 61. Panagoulias, D.P.; Virvou, M.; Tsihrintzis, G.A. Regulation and validation challenges in artificial intelligence-empowered healthcare applications—The case of blood-retrieved biomarkers. In Proceedings of the 14th International Joint Conference on Knowledge-Based Software Engineering (JCKBSE 2022), Larnaca, Cyprus, 22–24 August 2022; Volume 133.
- 62. Lundberg, S.M.; Lee, S.I. A unified approach to interpreting model predictions. Adv. Neural Inf. Process. Syst. 2017, 30, 4768–4777.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.